Author’s response to reviews

Title: Endomyocardial biopsy guided by intracardiac echocardiography as a key step in intracardiac mass diagnosis. A case report series.

Authors:
Marco Zanobini (marco.zanobini@ccfm.it)
Antonio Dello Russo (antonio.dellorusso@ccfm.it)
Matteo Saccocci (dr.saccocci@gmail.com)
Sergio Conti (sergioconti.md@ggmail.com)
Elisa De Camilli (elisa.decamilli@iieo.it)
Giulia Vettor (giulia.vettor@cccfm.it)
Valentina Catto (valentina.catto@cccfm.it)
Maurizio Roberto (maurizio.roberto@cccfm.it)
Cesare Fiorentini (cesare.fiorentini@ccccfm.it)
Giuseppe Viale (giuseppe.viale@iiieo.it)
Claudio Tondo (claudio.tondo@ccccfm.it)
Michela Casella (michela.casella@ccfm.it)

Version: 2 Date: 15 Jan 2018

Author’s response to reviews:

RESPONSE LETTER

Dear Editorial Office,

We have to proceed to modify the manuscript according all request from you and from reviewers.
1. DATA AVAILABILITY Modified as requested:
   - “Availability of data and materials: the datasets used and/or analyzed during the current study is available from the Corresponding Author on reasonable request.”

2. Please independently list each author AUTHORS’ CONTRIBUTION:
   - MZ and MC equally collaborate to the work
   - Data collection/analysis: MZ, ADR, MS, SC, EDC, GV, VC, MR, CF, GV, CT, MC
   - Manuscript designing/drafting/writing: MZ, MC, GV, VC, MS
   - Revision: MZ, MC, MS

3. AUTHORSHIP CRITERIA:
   - We confirm the authors’ list

4. ‘CONSENT TO PUBLISH’ modified as requested:
   - “extensive informed consent was signed by the patients at admission (anonymized clinical data and images)”

5. Manuscript submitted:
   - Final unmarked version

6. REVIEWER OBSERVATION
   - #reviewer 1: not applicable
   - #reviewer 2: language errors
   - page 6: "...eliminates"
   - page 6: "...surgery is reserved only for patients who would benefit the most..."
   - page 7: "...EMB can be performed..."
Best Regards

Matteo Saccocci, MD
Marco Zanobini, MD PhD
Michela Casella, MD PhD